Contents lists available at ScienceDirect

# Energy

journal homepage: www.elsevier.com/locate/energy

# The influence of socio-technical variables on vehicle-to-grid technology

# Shemin Sagaria<sup>a,\*</sup>, Mart van der Kam<sup>b</sup>, Tobias Boström<sup>a</sup>

<sup>a</sup> Renewable Energy Group, Department of Physics and Technology, UiT-The Arctic University of Norway, Tromsø, Norway
<sup>b</sup> Psychology of Sustainability and Behaviour Change, Faculty of Psychology, University of Basel, Switzerland

# A R T I C L E I N F O Handling editor: G Chicco

Vehicle-to-grid technology (V2G)

Keywords:

V2G acceptance

Charger power

EV adoption

EV battery availability

ABSTRACT

Vehicle-to-grid technology (V2G) is a novel large scale energy storage option to improve the grid integration of renewable energy sources (RES). Using electric vehicle (EV) batteries to store and provide electricity to the grid, the intermittency of RES can be reduced. However, a successful implementation of this technology depends on various social and technological factors. This study analyses the influence of social and technical factors such as V2G acceptance (percentage of EV users that allow V2G energy transfer), EV battery availability for V2G service (percentage of the EV battery), EV adoption level (number of EVs in EV fleet) and the charger power for the energy transfer between the storage system and grid.

Using Germany for the case study, a simulation model is developed and employed for the study. The base simulation results show that Germany needs 190 GW of PV and 170 GW of wind turbine installation to meet 80 % of electricity generation from RES in 2030. Further results show that an increased V2G acceptance and EV battery availability reduces the V2G contribution from individual EVs. A V2G acceptance of only 30 %, using half of the battery capacity dedicated to V2G, can help to reach an hourly reliability of 86.6 %. Having 50 % V2G acceptance, the hourly reliability increases by 5.3 %–91.9 %. The final analysis on charger power and EV adoption level highlights that a normal charger power of 7 kW or 11 kW can successfully accommodate V2G service. The study's overall results indicate that V2G technology will be an effective storage solution for Germany in the future.

#### Nomenclature

ESS	Energy storage system
EU	The European Union
EV	Electric vehicles
EV cons (t)	EV energy consumption
GW	Gigawatt
GWh	Giga watt-hour
HP	Hydropower system
HR	Hourly reliability
IEA	International energy agency
IRENA	International Renewable Energy Agency
kW	Kilo watt
kWh	Kilowatt-hour
L <sub>B</sub>	Load balance
LCOE	Levelised Cost of Energy
m <sub>share</sub>	EV share of moving vehicles
PHS	Pumped hydro storage system
P <sub>share</sub>	EV share of parked vehicles
PV	Photovoltaics
RES	Renewable energy source

(continued on next column)

\* Corresponding author.

E-mail address: shemin.sagaria@uit.no (S. Sagaria).

https://doi.org/10.1016/j.energy.2024.132299

Received 19 May 2023; Received in revised form 3 May 2024; Accepted 1 July 2024 Available online 4 July 2024

0360-5442/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).







S <sub>ava-EV</sub>	The storage capacity available in EV
SOC	State of Charge
SOC <sub>m</sub>	Updated SOC of moving vehicles
SOC <sub>max</sub>	Maximum SOC to which EV can store energy through V2G
SOC <sub>min</sub>	Minimum SOC to which EV can discharge through V2G
SOC <sub>P</sub>	Updated SOC of parked vehicles
SOC <sub>total</sub> (t)	State of charge of EV fleet at the time 't'
SOC <sub>temp</sub>	
Temporary SOC point betwee 't' and 't+1'	een time
SS	Self-sufficiency
TW	Tera watt
TWh	Tera watt-hour
V2G	Vehicle-to-grid technology
VRE	Variable renewable energy sources

## 1. Introduction

The development of society is dependent on energy. An absence of sufficient energy generation can affect the supply and demand of various sectors. Around the world, all countries intend to develop clean energy sources to reduce environmental impacts. Using alternate and renewable energy sources (RES) and conserving energy are the two aspects of sustainable development in energy systems. RES is being used instead of non-renewable resources, such as solar power, wind energy, and hydropower, all of which are low-emission and renewable. The European Union is aiming to reduce 40 percent of greenhouse gas emissions in 2030 as compared to 1990 levels [1]. Among other plans, the EU aims to increase the share of energy production from renewable sources by up to 32 percent by improving the efficiency of systems [2]. To achieve these goals, it is necessary to replace fossil energy sources with RES. Additionally, the European Union Commission presented its strategy for a prosperous, competitive, and climate-neutral economy by 2050 [1]. RES are a driving force for sustainable development in the energy sector [3]. As RES can generate energy with very little carbon emissions, RES can meet carbon emission standards. The limitless availability of renewable sources will be more than enough to meet the requirements of the world's needs [4]. Despite this, a lack of infrastructure and the intermittent energy generation are hindering the growth of RES. An important solution to balancing these issues is energy storage. In an energy storage system (ESS), the excess electricity from RES is stored and then used when energy production from RES is low [5].

Since the early 21st century, RES application has increased for electricity generation as an alternative to fossil energy sources. In the energy sector, Solar photovoltaics (PV), wind and other sources have been widely used for the generation of electricity and heat. Since then, several more studies focus on the development of an electricity grid with 100 % electricity generation from RES, supported by ESS. Hrnčić et al. [6] studied the possibility of achieving a 100 % RES in Montenegro. Their results from the EnergyPLAN model indicate a positive response on achieving 100 % RES grid with ESS. The study by Tetteh et al. [7] focuses on cost-effective electro-thermal ESS. The hypothesis focuses on modular electro-thermal energy storage (ETES) with various storage materials (thermal oil, molten salt, and sand) at high capacities and high efficiencies. The results show sand as an efficient storage medium with higher efficiency (85 %), with extremely high-temperature tolerance, increasing the Carnot efficiency of Stirling engines and driving down the cost of ESS six times compared to other technologies. Teki et al. [8] studied residential units with solar and ESS. The results show that the system performs well during daytime conditions, including peak shaving. Al-Ghussain et al. [9] conducted a study on the microgrid in a university campus with 100 % RES. According to the results of the study, the hybrid system guarantees high self-sufficiency, approaching almost 99 %. The study by Giarola et al. [10] analyses the role of ESS combined with RES. The study encompasses a model comparison approach where four models (GENeSYS-MOD, MUSE, NATEM, and urbs-MX) are used to analyse storage uptake in North America. The results show that storage may promote emission reduction at lower costs when renewable mandates are in place whereas, in the presence of carbon taxes, renewables may compete with other low-carbon options.

Despite these studies showing the feasibility of RES integrated with ESS for a 100 % RES-supported grid, the development towards 100 % RES systems in the real world has been slow. One of the reasons for this is that the development of ESS technologies requires additional investment in the form of money and resources. Simultaneously, money and resources are being invested within the transportation sector to electrify large parts of the sector, by replacing fossil cars with electric vehicles (EVs), gasoline stations with EV charging. Replacing fossil-fuelled vehicles with EVs, it is estimated that each vehicle can reduce its emissions by between 25 % and 80 % depending on the electricity source [9]. This electrification process and transition towards EVs provide a

great opportunity to develop ESS through Vehicle to Grid (V2G) technology with much less additional investments compared to developing large-scale thermal, mechanical or electro-chemical ESS. In V2G technology, the EV is connected to the grid through a bi-directional charger, which facilitates energy storage and extraction. Considering the only investments in bi-directional chargers instead of uni-directional chargers, the total financial and resource investment would be much lower compared to the development of the new ESS.

In our study, we focus on vehicle-to-grid technology (V2G), which is stored in EV batteries. Through bi-directional chargers, the technology offers a storage capacity of 20 MWh of energy from 200 EVs, considering 100 kWh batteries. The initial study by authors to support Germany with V2G as an ESS indicates that V2G can be a viable ESS [11]. Another study by Sagaria et al. [12] shows that Spain can achieve its 2030 and 2050 renewable energy goals with EV as the primary ESS through V2G, facilitating high penetration of RES into the grid. The study on V2G transfer and applications by Lakshmi et al. [13] points out various applications, such as vehicle-to-home and building energy transfer during peak load periods. Mozafar et al. [14] performed a study on smart grid operations considering large-scale integration of EV to grid through V2G technology. The results showed that an EV is a good ESS for the smart grid, which can virtually eliminate the need to use high-cost generators or other ESS to supply the system in peak hours ergo reducing the hourly cost of the system. Many other studies also support the integration of EVs into the grid to develop virtual ESS [15-18]. Further, Sagaria et al. [19] analyses photovoltaic integrated EVs, where EVs generate energy themselves through onboard PV and can transfer this energy to the grid through V2G. These EVs can be further utilized during peak load periods.

In a recent study by Zhao et al. [20], the environmental impacts of Vehicle-to-Grid (V2G) technology were evaluated within the context of a 2050 United Kingdom system, employing a consequential Life Cycle Assessment (LCA) methodology. The results show that the implementation of V2G has the potential to effectively offset the environmental footprint of electricity generation in high RE scenarios. Ramaiah et al. [21] study on the microgrid using fast charging DC architecture shows that V2G coupled with proper controller provides excellent dynamic performance in terms of dc bus voltage stability, and the charging station design assures low harmonic distortion of grid injected current. The study by Li et al. [22] introduces a scheduling optimization model for an integrated energy system with EV and hydrogen vehicles. Their results show that with V2G mode the total cost of power is reduced by 7.8 %, and the cost of power purchased from the grid decreases by 53.7 %.

Boström et al. [23] analyse a pure PV-EV energy system PV as the only electricity source working solely with EVs to satisfy the nationwide energy requirement in Spain. Their result showed that an hourly reliability of 100 % is possible with 73 m<sup>2</sup> of PV per capita in Spain, solely using EVs for energy storage and balancing. The study by O'Neill et al. [24] assesses the V2G operation in the microgrid to support Variable Renewable Energy (VRE) generation. The simulation results over 1 calendar year shows that V2G reduces the LCOE grid cost by 5.4 % for the solar supported microgrid, 4.6 % for wind supported micro grid and 4.5 % for solar and wind supported microgrids. Schuller et al. [25] developed an optimization model aimed at maximizing the utilization of EVs through V2G and VRE under various power generation and charging infrastructure scenarios. Their findings highlight the potential of coordinated charging, which could more than double VRE utilization, although the effectiveness is constrained by the length of the lookahead period. Nezamoddini and Wang [26] approached the challenge from the perspective of ISOs, incorporating uncertainties in VRE output, load and parking patterns, and transmission line reliability to optimize V2G dispatch. The studies [17,27-30] examine the peak load shaving capacity, techno-economic analysis and grid parameters using V2G technology. The results show that the power demand reduces by up to 6 % with a proper energy management system.

Whereas early studies demonstrated the high technical potential of V2G systems [31], more recent studies have started to focus on social factors of V2G systems that act as a barrier to widescale V2G deployment, such as user willingness to accept new technological developments and V2G acceptance. For example, a study by Esmaili et al. [32] indicates range anxiety as a key barrier to user participation in V2G. Range anxiety is the driver's fear that a vehicle has insufficient energy (battery capacity) to cover the distance needed to reach its intended destination, and would thus strand the vehicle's occupants mid-way. Geske and Schumann [33] analysed the willingness of EV drivers to participate in V2G technology and the different reasons for their decision. Their result shows that range anxiety and minimum range are the most determinant decision factors for V2G participation. Kester et al. [34] study variables promoting V2G in the Nordic region and also point out the lack of infrastructure, battery degradation and consumer awareness as main hurdles for V2G growth. Their study points to user concerns over their battery degradation through V2G purposes. Fast DC chargers over 50 kW, enable charging and discharge of batteries over 1C rate, which can increase chemical and thermal stress in the batteries, which reduces battery life. A study by Noel et al. [35] studied different barriers apart from range anxiety. Their study was based on a survey and results show that reasons for low V2G growth include low EV adoption levels, poor business models, and battery degradation. The diffusion study on PV and EV by Van der Kam et al. [36] shows the adoption of EV and its implication on the transition to smart grid systems. Focusing on the Netherlands, their study shows that EV adoption levels have a high influence on V2G potential.

From the former studies, we identify range anxiety, battery life anxiety, V2G acceptance and EV adoption level as key variables. Noel et al. [35] show that 28 % of the EV users who attended the interviews prefer other energy balancing technologies than V2G and 17 % show resistance due to the unwillingness to accept a third party to access their battery or the concept of V2G was too complex. While 12 % of the participants worry about battery degradation due to V2G and 10 % of the participants indicate V2G is not practical because of the lack of a large EV adoption level. A study by Meelen et al. [37] also mentions that user preference, technology and infrastructure all have a high impact on upscaling of V2G technology. Kester et al. [34] study variables promoting V2G in the Nordic region and also point out the main challenges for the faster development of the technology as battery degradation and consumer awareness. In this study, we aim to understand the potential impact of these variables through simulating scenarios with different V2G acceptance rates, battery capacity, charger power and EV adoption levels.

Even though past technical research, analyses the possibility of EV as ESS through V2G, they failed to demonstrate how the change in social behaviour affects the results. On the other hand, social behaviour studies demonstrate the influence of social variables in V2G technology without technical details. Our paper aims to combine both technical and social factors to generate insightful results. Multiple RES (Wind, PV, and hydropower systems) and ESS [EV and pumped hydro system (PHS)] are included in the model which helps to simulate real-life scenarios. We consider EV as the primary ESS and PHS as the secondary ESS to give more focus on the V2G energy flow. As secondary ESS, instead of PHS, we can include other ESS such as thermal and electrochemical ESS or multiple storage systems. Through this conceptual study, we look forward to delivering meaningful insights by analysing primarily four variables, 1) V2G acceptance rate - which illustrates how many EV users allow energy transfer between EV and grid through the centralized control system. 2) EV battery availability - what percentage of the battery can be used for V2G purposes and avoid range anxiety between users. 3) Charger power - to study whether fast chargers are necessary or not for successful V2G execution, and finally, 4) EV adoption level - to study how the change in the number of EVs influences the total outcome of the technology.

For better insights, the study focuses on Germany. We choose

Germany due to its diverse electricity generation portfolio [38]. In the latest update in Erneuerbare Energien Gesetz (EEG - The Renewable Energy Act), Germany aims to generate 80 % of their electricity from RES in 2030 and increase the share of wind and PV to 115 GW and 230 GW respectively [39]. Considering Germany's 80 % electricity generation milestone in 2030, it is vital for a RE-supported grid system to have ESS. Building new ESS requires more resources and infrastructure. Considering the estimates of 15 million EVs in 2030 [40], this provides a platform to test the influence of V2G technology in the grid system, where we can save more resources and emissions from those infrastructures by using EVs as ESS. Our model does presently not include electricity imports and exports. Germany is a country with many neighbours and, electricity exchange with them is obvious and generates better energy security. Electricity imports and exports will be added in later versions of our model.

The research paper is structured as follows, Section 2 details the development of the simulation model. Section 3 explains the operation condition of the model and the initial setting. Section 4 gives the results from the simulation model under different operation scenarios. Section 5 discusses the results and outcomes in detail and section 6 presents the conclusion of the study.

## 2. Model development

To study the influence of social variables in V2G, a simulation model is developed in MATLAB/Simulink software. Previous studies in the field focus solely on one RES or ESS technology or a combination, and its influence on the grid. In the real world, multiple RES is integrated with ESS to satisfy energy needs. This underestimates the flexibility within the energy systems. Our study aims to fill this void through a model that considers multiple energy sources and storage systems for the analysis and includes the variables to perform this study.

The primary goal of the study is to perform an analysis of the influence of socio-economic variables on V2G technology. In the model, we consider a top-down analysis approach to study the influence of variables on V2G electricity flow. Since V2G technology is still in the development phase and with little research on the V2G acceptance rate and average expected EV battery availability, it requires further largescale pilot experiments and surveys. Additionally, there are presently not enough EV volumes and chargers for V2G purposes (considering 1:1 bi-directional chargers). Because of these reasons, our study took a topdown approach, where we analyse scenarios with different V2G acceptance rates, EV battery availability, charger power and EV adoption levels.

In this study, PV, wind, and hydropower (HP) are considered as RES, and PHS and EV are considered ESS. The inputs are given in the load profile subsystem, EV subsystem and Operation subsystem. The load profile determines the hourly load requirement from yearly consumption. Based on the electricity consumption profile from TSO/DSO, the yearly consumption is segregated into hourly load profiles. This gives the electricity consumption from the household and the industry each hour in a year. Also, the subsystem estimates the electricity generation from RES. The estimations are based on the yearly electricity generation profile from Germany, obtained through the Agora Energiewende GmbH database [21]. The vehicle subsystem estimates the EV battery capacity from the EV for the V2G service. The EV fleet characteristics like usage profile for each day, number of EVs, and battery capacity are input to this subsystem. The operation subsystem derives the electricity flow between RES and ESS to give the required outputs. Fig. 1 shows the model developed in Simulink software.

Germany's annual average electricity consumption is approximately 520 TWh, consumed for transport, trade and service, household, and industrial applications [41]. Based on the user profiles of the household and industrial section from Stromnetz Berlin [42], the hourly load profile is generated. For RES, since solar and wind installations are growing rapidly, the simulation uses annual electricity generation data



Fig. 1. Model A structure in Simulink.

from 2020. At the end of 2020, Germany had a PV installed capacity of 54 GW and 62.67 GW of wind energy combining both onshore and offshore installation. In 2020, a combined 181 TWh of electricity was produced from both PV and wind (50.7 TWh from PV and 130.32 TWh from wind). From this, the hourly generation for 1 MWh is calculated for a year. This is further used. For Hydroelectric power, the electricity generation profile from 2020 is given as the input. In 2020, 18.3 TWh of electricity was produced through hydropower in Germany [21].

The model considers a total load of 520 TWh/year for Germany. We assume the electricity requirements will be the same even in 2030. The past electricity consumption data from 2000 to 2021 [41], shows the change in electricity consumption as  $\pm$ 40 TWh from the average of 500 TWh. From 2010 to 2021, the yearly electricity consumption is decreasing at a rate of 7 %. In our analysis, we consider EVs and they can increase the electricity requirement in the future. However, the model

calculates EV energy requirements separately in addition to the electricity load. This assumes that the electricity requirement in 2030 is valid, nevertheless, extrapolating historic consumption data results in a reduction of total electricity consumption in 2030 as compared to the assumption.

In this study, we investigate the maximum potential of V2G for grid balancing. Therefore, the model considers the EV battery as the primary ESS and PHS as the secondary ESS. For the simulation, the average driving distance per day is 49 km, with an average energy consumption of 200 Wh/km [23,43]. For the V2G process, the round-trip efficiency of the V2G technology is 87 % with 93.5 % efficiency for charging and discharging respectively [23]. In the base case study, the maximum and minimum state of charge (SOC) of the EV fleet is confined between 90 % and 10 % to avoid extra energy losses, battery degradation and improved battery life [44]. The EV driving distribution represents the



Fig. 2. EV driving distribution.

vehicle moving pattern each hour for a day. The EV driving distribution is generated through a log-normal function, showing the percentage of vehicles moving each hour [23]. Fig. 2 shows the moving distribution of EVs for a day. We assume the same pattern will be followed all day around the year. The energy consumed by the EV is calculated individually using (1).

percentage of EVs driving at timestep t.

The model considers PHS as the secondary storage system, to focus more on energy flow through EVs through V2G technology. Germany currently has a PHS capacity of 5355 MW [45]. For this study, 5355 MW capacity is considered with 85 % round trip efficiency, assuming the state-of-the-art technologies employed. Fig. 3 represents the logical flow diagram of the model. The minimum and maximum SOC for EV and PHS, initial SOC and energy profiles were given as inputs for the model.

At the start of the simulation, the model calculates the load balance in the system, through equation (2). The load balance is the difference between the load requirements and the electricity provided by PV and wind installations. The load balance can be positive, zero or negative. A zero-load balance indicates energy generation from RES equal to the load demand. The positive load balance represents higher electricity consumption than electricity generation from RES, and the negative load balance represents less electricity consumption than electricity generation from RES.

Load balance 
$$(L_B)(t) = (\text{load cons.}(t) + \text{EV cons.}(t))$$
  
- (Wind energy  $(t) + \text{PV}(t)$ ) (2)

Electricity generation from RES is insufficient to meet the load demand during a positive load balance. Electricity from external sources is necessary to balance the requirements. In this model, the external source is ESS. The model analyses the SOC of both ESSs. If electricity from the EV battery is enough to satisfy the additional requirement, the electricity is extracted from the EV. If more electricity is needed, the rest will be taken from PHS after the EV batteries successfully discharge to the minimum SOC. The hydropower system activates when the PHS electricity supply is insufficient to meet the load demand. If there is still an electricity demand, it is considered an energy shortage point where the required electricity must be supplied by external sources.

A negative load balance represents excess electricity generation from RES over the electricity demand for that hour. The SOC of the EV fleet is checked at the beginning. If the SOC is less than the maximum SOC, electricity is stored in the EV batteries until the SOC reaches the maximum. If more electricity is still available, it is stored in the PHS system. If electricity is available even after storage, it is considered excess energy. This energy is wasted or discarded without being used on the grid.

The SOC of the EV fleet during each period is calculated as the sum of the SOC of the parked vehicles  $(SOC_p)$  and moving vehicles  $(SOC_m)$ , as in equation (3). At the beginning of each time step, the model assumes the same SOC for all the vehicles. The  $SOC_m$  of EV includes the energy required to drive during the hour, thus capturing the dynamic nature of the EV fleet. Based on the driving distribution a fraction of vehicles will not be able to connect to the grid. This proportion of vehicles uses their battery only for transportation purposes, which is the reason for estimating the SOC of moving vehicles separately. While it gets parked, the vehicle is connected to the grid and only the allowed battery limits for parked vehicles can be used for V2G purposes, not the whole vehicle



(1)

Fig. 3. Flow-diagram of the simulation model.

(4)

9

fleet. The SOC of the parked vehicles and moving vehicles is estimated individually using equations (4) and (5).

$$SOC_{total}(t) = p_{share}(t) \times SOC_p(t) + m_{share}(t) \times SOC_m(t)$$
 Equation 3

$$m_{share}(t) = EV driving distribution(t) \times total EV fleet capacity$$
 (8)

The SOC of PHS is also estimated each period through equation (9). Similar to the EV fleet, the SOC estimation depends upon factors such as the load balance and energy flow with the EV.  $SOC_{temp}$  is used as the intermediate point to determine the electricity flow in PHS. Case (a)

$$SOC_{p}(t) = \begin{cases} a) \text{ if } SOC_{temp}(t) > p_{share}(t) \times SOC_{max}; p_{share}(t) \times SOC_{max} \\ b) \text{ if } SOC_{temp}(t) < p_{share}(t) \times SOC_{min}; p_{share}(t) \times SOC_{min} \\ c) \text{ if } SOC_{remp}(t) < p_{share}(t) \times SOC_{max} \delta SOC_{temp}(t) > p_{share}(t) \times SOC_{min}; SOC_{temp}(t) \end{cases}$$

$$SOC_m(t) = SOC_m(t) - EV cons(t)/total EV fleet capacity$$
 Equation 5

Where  $p_{share}$  is the share of SOC of the parked vehicle and  $m_{share}$  is the share of SOC of the moving vehicles at the time step 't',  $SOC_{temp}$  is the temporary SOC of the EV fleets used for calculation purposes,  $SOC_{max}$  is the maximum SOC attainable by the EV fleet and  $SOC_{min}$  is the minimum SOC attainable by the EV.  $SOC_p$  calculation considers 3 operational cases. Case a) represents when there is excess energy generation. In this case, the EV recharges the battery to  $SOC_{max}$ . Case b) represents when an EV does not have enough electricity to meet the required loads. In this case, the EV discharges until EV reaches the  $SOC_{min}$  state. Case c) represents when EV can accommodate the whole excess energy generation from RES.

 $SOC_{temp}$  is a variable used to calculate the intermediate SOC of the moving and stationary EV. When we have excess electricity from RES or insufficient electricity from RES, the energy flow happens between the parked EV vehicles. Due to this, the SOC estimation of the parked vehicle must consider 3 different cases, positive load balance with enough electricity to meet additional requirements (case c), positive load balance with not enough electricity to meet the requirements (case b) and negative load balance (case a). Equation (6) shows the calculation of  $SOC_{temp}$ ,

represents additional electricity availability after charging the EV fleet to  $SOC_{max}$ . The excess electricity is stored in the PHS system and  $SOC_{PHS}$  is updated. Even after electricity is available after PHS reaches PHS-max<sub>SOC</sub>, the available electricity is discarded. Case (b) and (c) represent positive load balance points. When EV cannot fulfil the excess load requirements, energy from PHS is extracted, represented by the case (b) and If PHS cannot fulfil the additional requirement, the time period is marked as an energy shortage point. case (c) shows when EV stores the energy excess electricity from RES.

$$E \text{ bal.}_{PHS} = \begin{cases} a) SOC_{temp} > SOC_{max}; SOC_{temp} - SOC_{max} \\ b) SOC_{temp} < SOC_{min}; SOC_{temp} - SOC_{min} \\ c) SOC_{temp} < SOC_{max} \delta SOC_{temp,n} > SOC_{min}; 0 \end{cases}$$
Equation

Finally, the system's self-sufficiency (SS) and hourly reliability (HR) of the system are calculated. Hourly reliability indicates whether the RES and ESS system can satisfy the electricity requirement for each hour without external support. At the same time, system SS indicates the amount of electricity provided to meet the requirements over a year of RES and ESS systems. Equations (10) and (11) represent the calculation of HR and system SS.

$$HR = \sum_{1}^{8760} \left\{ E \ bal._{PHS} + E_{HP} < 0; 0 \\ E \ bal._{PHS} + E_{HP} \ge 0; 1 \right\} / 8760$$
(10)

$$SOC_{temp}(t) = \begin{cases} a \text{ if } L_B(t) < 0; p_{share}(t) \times SOC_p(t) + L_B(t) \times charging \ eff \\ b \text{ if } L_B(t) > 0 \ \delta \ p_{share}(t) \times SOC_p(t) > L_B(t); p_{share}(t) \times SOC_p(t) + \frac{load \ balance(t)}{charging \ eff} \\ c \text{ if } L_B(t) > 0 \ \delta \ p_{share}(t) \times SOC_p(t) < L_B(t); p_{share}(t) \times SOC_p(t) \times charging \ eff \end{cases}$$
(6)

 $SS = \frac{Total \ energy \ supplied \ by \ RES \ \& \ ESS}{Total \ energy \ required} \times 100$  Equation 11

3. Application of the model

The  $SOC_{temp}$  calculation varies according to the load balance. During the negative load balance, the electricity is stored in the EV, as shown in case (a). The inflow of electricity to EV revises  $SOC_{temp}$  updated with a positive number. During positive load balance, the model checks the capability of the EV to satisfy the additional load requirements. Case (b) represents when EVs can meet the requirements through energy discharge from the EV fleet. Case (c) represents a scenario in which the EV cannot meet the total additional load requirement. energy discharge from the EV fleet, where EVs cannot meet the additional electricity requirement. Equation (7) and 8 show the equation to calculate  $p_{share}$ and  $m_{share}$ , respectively.

$$p_{share}(t) = (1 - EV \text{ driving distribution}(t)) \times \text{total EV fleet capacity}$$

Equation 7

The goal of the model is to facilitate a study to analyse the influence of social parameters such as V2G acceptance, charger power, EV adoption and EV battery availability on V2G service. Before the analysis, the initial step is to determine the base state. Base state refers to the results of simulations of a particular scenario. In this study, we establish the base state based on the 2030 energy goals of Germany. In 2030, Germany is aiming to have 80 % of its electricity from RES, primarily solar and wind sources [46]. In addition, Germany estimates that there are 15 million EVs on the road [40]. Considering these statistics and assuming instant energy transfer from EV to the grid and 100 % V2G acceptance,



Fig. 4. Hourly electricity mix of Germany for 2 different days.



Fig. 5. RE installation and excess electricity generation for different self-sufficiency points.

an initial simulation will give the base results for further studies.

Using the developed model, estimating the amount of RES installations needed to reach 2030 requires multiple simulations. During each simulation for RES installations, the model predicts excess energy, energy shortages, system SS, HR and SOC of different ESS. Using the model, we will estimate PV and wind installations for the operation condition with different systems SS and HR. Analysing different SS levels through the model helps to identify trade-offs between PV-wind installations and other sources. A system with 100 % SS must meet all electricity needs at any particular time, but this can result in excess electricity generation. By reducing SS, it is possible to decrease the amount of RES installation required and the amount of excess electricity generated. However, other sources should be used to compensate for this decreased capacity of PV-wind installations. Based on the current energy mix and future forecasts, biomass might be an attractive option. This is further discussed in Section 4.1 (Fig. 5 shows the RES installation requirements for different SS). In 2030, the goal of Germany is to generate 80 % of its electricity from RES. In the simulation, we only consider PV, wind, and hydropower. Nevertheless, we can see from the electricity mix chart of Germany (Fig. 4) that it has other RES such as biomass and other RES contributions [47].

Currently, biomass generates around 10 % of the total electricity from RES in a day. The study by the International Renewable Energy Agency estimates that 90 % of total electricity generation in Germany for 2030 will be from PV, wind and hydro and the rest from biomass [48]. Considering the biomass contribution, PV and wind contribution are estimated to be 72 % (90 % of 80 %) of the total electricity mix in 2030. Along with this, the final 20 % of the electricity is assumed to be delivered from other non-RES. Based on the electricity mix profile of Germany, illustrated in Fig. 4 (note that the solar and wind resources change from day to day and with the season) we can observe that non-renewable electricity forms the base load. Base load represents the constant output from the source over the period. Even though the fact that gas power plants offer flexibility, in this study we treat all non-renewable sources as non-flexible to primarily study the influence and contribution of V2G on the grid. Hence, we assume that 20 % of electricity generated by non-RES sources is the base load.

After defining the base case, the simulation model is used for the analysis of social and technical variables. The parametric study includes i) an Analysis of V2G acceptance vs EV battery availability and ii) an Analysis of Charger power vs EV adoption level. Considering the top-down approach, where we analyse scenarios with different V2G acceptance rates from 100 % to 0 %, EV battery availability from 80 % to 20 %, charger power from slow AC charger (2.3 kW) to fast DC chargers (100 kW) and EV adoption level (from 1.8 million to 35 million EV in 2030) to get insights over various development stages. The variables in

the study offer 6 different options to combine for the analysis. Here we present the results from the analysis of V2G acceptance vs EV battery availability and charger power vs EV adoption level. The appendix section presents additional results, which are i) analysis of V2G acceptance vs charger power, ii) analysis of V2G acceptance vs EV adoption level, iii) analysis of EV battery availability vs charger power and iv) analysis of EV battery availability vs EV adoption level. Finally, a sensitive analysis analyses how the change in assumed SS affects the results.

#### 4. Simulation and results

Through a top-down analysis approach, the study progresses through 4 steps using the simulation model in a MATLAB/Simulink interface. Sections 4.1 to 4.4 provide the results and details of the simulation. In addition to this, from the simulation. In addition to this, Appendix Fig. 1 - Appendix Fig. 4 shows further results from the cross-analysis of all four variables.

# 4.1. Renewable energy installation analysis to satisfy the 2030 energy goals (base case study)

We estimate the requirements of the RES system to meet the 2030 goals by running multiple simulations. As mentioned in Section 3, it would be difficult to determine the required installation directly with a single simulation. Consequently, multiple simulations are done with the model for a different combination of RES. At the end of 2020, Germany had an installed PV capacity of 53.8 GW and 62.7 GW of wind energy, combining both onshore and offshore installations [49,50]. Using the present installation as a reference state, we can estimate the HR for different combinations of PV and wind installations. For SS between 90 % and 100 %, the initial simulation model then determines the required wind installation to meet the electricity demands. Fig. 5 shows the results obtained from the simulations.

From Fig. 5 we can observe that to meet 100 % SS compared to an SS of 99 %, an additional 90 GW of wind energy installation is needed with 190 GW of PV installation. It is due that in winter, the specific hourly demand can be very high, compared to the average hourly demand. Installing more RES can satisfy the demand, but this higher installation generates more electricity in the summer time than the demand which increases the excess energy generation as shown in Fig. 5. By relaxing the SS target, the required RES installation and the resulting excess electricity are decreased. Due to the intermittent nature of wind and solar energy, there is a trade-off between increasing their installed capacity to reach SS goals and reducing excess electricity. Increasing other RES, such as biomass, reduces energy system intermittency, allowing to achieve higher SS without increasing excess energy. We explore this trade-off below and argue for a 99 % SS target as a reasonable mid-point in this trade-off.

For a 100 % SS system with 190 GW of PV, 255 GW of wind installations and 288 TWh excess energy generation, PV-Wind contribution is 72 % of the total electricity, while other RES, mainly biomass contributes 8 % and non-RES contributes the rest of the 20 %. Reducing the SS target to 99 % reduces the required wind installation to 170 GW with 190 GW PV, generation of 121 TWh excess energy. Reducing SS further to 95 % and 90 % reduces wind installations further down to 135 GW and 114 GW respectively with excess energy generation of 59 TWh and 30 TWh. However, decreasing installed wind and solar capacity

Table 1	
Simulation results from 190 GW of PV and 170 GW of wind energy installations	5.

PV installation (GW)	Wind energy installation (GW)	Hourly reliability	System self- sufficiency	Excess energy (TWh)	Energy shortage (TWh)
190	170	98.87	99	121.2	3.7

needs to be compensated by increased use of other sources.

Here, we focus on biomass as a replacement energy source, which is planned to contribute 10 % of renewable electricity in 2030 in Germany [48]. For a 99 % SS target based on wind and solar energy, the electricity contribution from PV and wind reduces by 0.72 % and reduces the final contribution from 72 % to 71.28 %. Using bioenergy to cover this small fraction increases the bioenergy share from 8 % to 8.72 % in annual electricity generation, which requires an additional land area of 9 %. While for SS targets of 95 % and 90 %, the share from biomass increases from 8 % to 11.6 % and 15.2 %, respectively. Also, this demands an additional land area of 45 % and 90 % for 95 % SS and 90 % SS respectively from bioenergy which seems unreasonably high compared to the current plans. Considering the reduction of wind installation from 255 GW (100 % SS) to 170 GW (99 % SS), 135 GW (95 % SS) and 114 GW (90 % SS) against the increase of land area by 9 %, 45 % and 90 %, we argue that 99 % SS point is a good compromise between reducing the wind and solar capacity target and increasing land-use for biomass. To explore the impact of setting other SS targets, we perform sensitivity analysis is performed for different SS targets (Section 4.4). Appendix Table 1 shows the complete results of the simulation.

From the 99 % SS scenario, we choose 190 GW of PV installation and 170 GW of wind installation, which results in the least excess energy generation of 128 TWh. For the initial simulation, we assume the most favourable condition towards V2G technology with 100 % V2G acceptance, an 11-kW AC charger for 15 million EVs with 80 % EV battery availability [44]. The PHS capacity is kept constant throughout the studies at 5.35 GW and the maximum available energy through the HP system is 18.38 TWh. Table 1 shows simulation results with 190 GW of PV and 170 GW of wind installations.

The simulation results in Table 1 show that to meet 71.28 % of the total electricity demand from RES, we require the installation of 190 GW of solar and 170 GW of wind. Additionally, we can observe that HR and system SS are both 99 %. This is because the simulation focuses on satisfying 99 % of the requirements instead of 100 %, as explained earlier. The results also show excess energy generation of 121 TWh. This excess energy is considered a waste because the ESS is at its maximum storage capacity. This can be reduced by increasing the ESS capacity or by exporting the energy to neighbouring countries. Increasing the ESS capacity also helps to reduce the installation required. This is further discussed in sessions 4.3 and 5.

#### 4.2. Influence of V2G acceptance and EV battery availability on the grid

The potential of V2G technology highly depends on social factors, mostly on the user acceptance of the V2G technology. V2G acceptance refers to the willingness of EV users to participate in V2G services. According to previous studies [31,33–35], several hurdles need to be overcome in order for V2G to become more widely accepted. According to previous research, such as Esmaili et al. [32], and Noel et al. [35], range anxiety is one other main barrier to V2G. Essentially, range anxiety is the fear that a vehicle does not have enough energy (battery capacity) to travel the distance required to reach its destination, leaving its occupants stranded. This range anxiety arises because of not limiting the battery available for the V2G service. With this analysis, we combine and study the two variables V2G acceptance and EV battery availability to understand how they influence the hourly dependence and energy flow through EV and RES-supported grids.

In the study, we simulated different V2G acceptance rates, from 0 to 100 %, for different battery capacities reserved for V2G purposes. For the simulation, we assume a 190 GW PV installation and 170 GW wind energy installation with 15 million EVs connected to the grid through an 11-kW bi-directional charger. Fig. 6 shows the results of the simulation. To show the change in HR for different scenarios more clearly, we plot HR results from 65 % (on the Y axis).

From the results in Fig. 6a, we can observe that the HR on the RESsupported grid is declining as the V2G acceptance rate and EV battery



**Fig. 6.** Influence of V2G acceptance and EV battery availability on hourly reliability and EV energy flow. Note: BA – EV battery availability; For 0 % V2G acceptance, the EV energy flow and charging cycle for V2G is 0.

availability drop. With a lower V2G acceptance rate, fewer EVs are connected to the grid for energy transfer. As a result of the reduction of EVs connected for V2G service, the total ESS capacity available through V2G is lowered. This will reduce the storage potential of energy. During periods with high-RES excess energy generation, total stored energy reduces and will decrease energy discharge during high-load periods. This reduces the SS and the HR of the system. While for the same V2G acceptance rate, a decrease in available battery capacity for V2G service also reduces the storage capacity through V2G service. Similar to the former scenarios, lowering EV EV battery availability will result in the reduction of SS and HR in the system. From Fig. 6a, it can be seen how the V2G acceptance rate and the EV battery availability finally affect the total energy storage potential through V2G.

Fig. 6b shows the change in energy flow through EVs and the number of charging dedicated for V2G purposes for different operational scenarios. From the results, we can observe that as V2G acceptance or EV EV battery availability reduces, the energy flow through EVs is also reduced. It is because of the reason mentioned before, the change in total energy storage capacity. This will then reduce the maximum possible storage/extraction, thus the energy flow. However, looking into the number of charging cycles each vehicle must go through, for different V2G acceptance rates and EV EV battery availability, we can observe that a low V2G acceptance rate puts more pressure on each vehicle. With 10 % V2G acceptance and 50 % EV EV battery availability, each EV has to go through 128 full charging cycles as compared to 76 and 58 charging cycles with 30 % and 50 % V2G acceptance.

From the results, we can observe that with 50 % V2G acceptance and 50 % EV battery availability, we lose approximately 7 % HR (to 91.9 %) as compared to the scenario with 100 % V2G acceptance and 80 % EV battery availability (from 99.02 %). The results also show a 30 % V2G acceptance rate with 50 % EV battery availability helps to achieve 86.6 % HR with 92.4 % system SS. Subsequently, the study indicates that V2G services using EV batteries can reduce RES intermittency issues, also if only less than half of the owners are willing to be part of the V2G service. Appendix Fig. 1 - Appendix Fig. 4 shows further results from the cross-analysis of the V2G acceptance rate and EV battery availability with



Fig. 7. Influence of V2G acceptance and EV battery availability on hourly reliability and EV energy flow. (Note: The results from 11 kW, 22 kW, 50 kW and 100 kW chargers are the same and overlap each other).

## other variables.

#### 4.3. Influence of charger power and EV adoption level on the grid

As the transportation sector electrifies rapidly, more EVs are being sold each year. By 2030, Germany aims to have 15 million EVs on the roads. As the number of EVs increases, it can have a positive impact on the energy storage potential. However, the expectation for the number of EVs can change. Our study assumes 15 million EVs to analyse different scenarios. Along with this, the EV charger power has a significant impact on its energy transfer to the grid. High-power DC chargers can transfer up to 350 kW, while regular home or work AC chargers normally peak at 11 kW. However, charging especially high-power charging can reduce battery life due to the thermal stress induced inside the battery, which is one of the concerns with V2G technology. In this analysis, we investigate the energy storage impact of the V2G technology on changes in the EV adoption level and charger power. To examine the energy flow and HR in the study, simulations were performed with different charger powers and EV fleet volumes. The study analyses the V2G technology using EVs from 1.4 million (current EV German fleet) to 35 million EVs (230 % of projection in 2030) and charger powers from 2.3 kW up to 100 kW. In this simulation, we consider 50 % EV battery availability for the V2G service, 50 % V2G acceptance rates, 190 GW of PV and 170 GW of wind turbine installations. Fig. 7 shows the results of the simulations.

In Fig. 7a and b, we can observe that the HR varies based on the EV adoption level and charger power. The results show that charger power with a maximum charging rate higher than 11 kW does not further increase the V2G potential. In Fig. 7a, we can observe that the HR increases as the number of vehicles increases up to 10 million EVs and reduces for higher EV adoption. This is because as the number of EVs increases, the electric energy required for the transportation sector and thus the total electricity demand increases. At the same time, an increasing amount of EVs increase the capacity to store excess electricity, which otherwise would have been wasted. The increased storage improves the HR and SS, during high-load periods. However, when the EV adoption level increases beyond 10 million, the extra storage capacity cannot be optimally utilized, because the RES output is constant. An EV fleet of more than 10 million leads to higher energy demand, reducing the HR and SS.

Fig. 7b shows the energy flow through the EVs and the estimated charging cycles from each EV. From the results, we can observe that as the EV adoption level increases, the energy flow through the EVs is also increasing. More EVs offer more space for excess energy storage and increased usage of stored EV battery energy, during low RES-energy

generation periods. Considering the results in Fig. 7b with a charger power of 11 kWh, we can observe that when increasing the EVs from 1.4 million to 15 million, the EV energy flow increases from 8.8 TWh to 43.6 TWh, which is about 5 times. While increasing EVs from 15 to 35 million EVs, the increased energy flow is only 11 TWh, up to 55 TWh. This reduced increase in energy flow reflects the reduction of HR. Also from Fig. 7b, we can observe that as the number of EVs increases, the number of charging cycles each vehicle has to go through is reduced. It is because as a greater number of EVs are available for V2G service, the energy flow is distributed between more vehicles. This reduces the number of charging cycles each vehicle must go through in a year.

While examining the influence of charger power, from the results in Fig. 7a and b, we can observe that increasing the bidirectional ACcharger power from 2.7 kW to 11 kW helps to improve the HR of the system, along with the EV energy flow. As the charger power increases, the ability of EVs to push energy to the grid increases. Assuming that the grid has a specific energy requirement (e.g., 1 MWh), the maximum energy contribution from each EV is limited by the ratio of total energy required to the total number of EVs participating in V2G. For instance, with 100 EVs participating in V2G, each EV would need to deliver an energy amount of 10 kWh. Chargers with lower power can only discharge up to their respective maximum power output (e.g., a 2.3 kW and 7 kW charger can only discharge at 2.3 kW and 7 kW, respectively), whereas chargers with higher capacities, such as 11 kW, 22 kW, or 50 kW, will only discharge the required 10 kW from each EV. In addition to this, the charger power also restricts the maximum energy flow from the grid to the battery which limits the total energy stored in the battery. Considering 1-year simulation, the results suggest that there is no need to have any DC high-power bidirectional charger, as there is no improvement in HR. Even the results for 7-kW chargers are almost on par with the 11-kW case. Consequently, a 7-11 kW charger would be sufficient to deliver the required energy back to the grid during the energy shortage period, for the considered scenario. This also implies that high-power DC chargers are not required to perform V2G services. Appendix Fig. 1 - Appendix Fig. 4 shows further results from the crossanalysis of the V2G acceptance rate and EV battery availability with other variables.

# 4.4. Sensitivity analysis on self-sufficiency vs V2G acceptance and EV adoption level $% \mathcal{C}(\mathcal{C})$

100 150 100 150 90 135 90 135 80 120 80 120 Energy flow through EV (TWh) through EV (TWh) V2G 120 70 105 70 105 Charging cycles/year for Charging cycles/year for 60 90 60 90 50 75 50 75 flow 40 60 40 60 Energy 30 30 45 45 20 30 20 30 10 15 10 15 0 0 20 25 35 0 5 10 15 30 0 10 50 100 20 30 40 60 70 80 90 of EV (in millions) Numb V2G acceptance rate (%) Energy flow SS - 99% Energy flow -SS - 100% SS - 99.99% SS - 97% SS - 100% -SS - 99.99% -SS - 99% SS - 97% Charging cycles ---- Charging cycles -SS - 95% -SS - 92% -SS - 90% SS - 95% SS - 92% -SS - 90% (b)(a)

For the former analysis, we have assume that the SS of the system is 99 %. Replacing fossil sources with RE sources to generate 80 % of electricity in 2030, 72 % of total electricity generation is expected to

Fig. 8. Sensitivity Analysis on Self-sufficiency vs V2G Acceptance and EV adoption level.

deliver by PV and wind [48]. With 99 % SS, the total generation would be 71.3 % instead of 72 % and the reduction is covered by other RE sources such as biomass to meet 80 % SS. However, in this section, we analyse the impact of SS on the system.

For SS ranging between 100 % and 90 %, the corresponding RE installation changes, as shown in Fig. 5. The RE combination with the least excess electricity generation points is chosen for further study. Considering EV battery availability of 50 % for V2G service and 11 kWh charger power, the study examines the change in HR for different V2G acceptance rates and EV adoption levels. Fig. 8 shows the result of the simulation.

As we can see from the results, the SS of the system has a noticeable impact on the simulation results. As SS decreases from 100 % to 99 % for 50 % V2G acceptance, EV energy flow and charging cycles increase by 40 % (Fig. 8a). This is because lowering SS reduces the installation of RE. A reduction in RE installation reduces electricity generation over hourly demand. At the 100 % SS point, the additional RE installation supplies more electricity, while the ESS contributes a small proportion to reach 100 % HR. While at 99 % SS, the EVs provide more electricity via V2G, to meet the hourly electricity requirements. Because EVs contribute comparably little in the 100 % case, only a small portion of additional electricity gets stored, and the rest goes unused (excess energy generation). With 100 % SS, RE sources generate 252 TWh of excess electricity, while 99 % SS generates less than half of that,121 TWh. The sensitivity analysis on EV adoption level also shows a similar behaviour, where reducing the SS increases the electricity flow through EVs and charging cycles/EVs for a year (Fig. 8b).

#### 5. Discussion

This research article presents a conceptual study which analyses the social and technical parameters of V2G technology. The study focuses on analysing the influence of variables such as V2G acceptance, EV battery availability, charger power and EV adoption level on V2G service and energy flow. Giving more attention to V2G services, the results from this study help to further understand V2G and its importance in large-scale RES integrated grid systems. The case study is based on Germany and the analysis is done to meet 2030 energy goals, to generate 80 % of the electricity from RES.

The results from the study are presented in three sections. In section one, we focused on the base case, which helped to identify the RES system required to meet the goals with best-case scenarios. Assuming the most favourable scenario for the base case with 100 % V2G acceptance rate, 11 kWh bi-directional chargers, 15 million EVs and 80 % EV battery availability for the V2G service. But this is practically impossible as V2G acceptance and EV battery availability for V2G service highly depend on EV owners and their approach towards the technology [31, 35,51,52]. To what extent EV drivers will participate in V2G in the future is highly uncertain. Next to consumer behaviour, EV policies and technological developments will influence further other relevant factors such as EV diffusion and charger power. The result of this study presents how these variables and their uncertainty influence the HR and SS of the grid and the energy flow between the grid and the EVs. A very important parameter that determines the EV user participation level is the economic incentive the car owner gets. This must be investigated using a questionnaire and will be performed in the coming research.

The initial results show that with favourable conditions, we can achieve the 2030 goals with 190 GW of PV and 170 GW of wind turbines, coupled with V2G for energy storage in EVs. This is approximately 3.7 times PV and 2.7 times wind turbines as compared to 2020 installations. These proposed installations for PV are less than Germany's PV estimation in 2030 by 40 GW, while wind turbines are lacking by 25 GW [53]. Considering the 2030 goal of Germany to have 230 GW of PV and 115 GW of wind, the simulation shows that Germany can reach SS between 92 % and 95 % (Fig. 5). With 230 GW of PV and 115 GW of wind installation, the total RE share in the 2030 electricity mix would be 75

%, including the 10 % contribution from biomass. However, it can be different in the real world, as Germany plans to diversify and improve its energy storage capacity to support the announced RES installation. While the simulation only considers the existing PHS system with EV. From the sensitivity analysis, we find that the trend followed is the same for different SS points. With 100 % or 90 % SS, the trend characteristics would be the same as those of 99 %, which is considered for variable analysis. However, between 100 % SS and 99 % SS, there is a 40%-50 % difference in energy flow, whereas from 99 % SS to 95 % the difference is 10 %. This is because high RE installations reduce EV energy contribution during higher SS points. Nevertheless, the projected 230 GW installation of PV in 2030 in addition to other supplementary RES would be sufficient to deliver the electricity. Having more RE installations improves the system's SS but also increases the excess energy generation and severely increases the use of material resources. In reality, the excess energy/electricity could be used for cross-border energy trading or in other innovative technologies such as hydrogen production.

In section two of the results, we present the impact of V2G acceptance and EV battery availability on V2G energy flow and grid HR. The results in Section 4.2 show that a reduction in V2G acceptance and EV battery availability has a negative influence on the grid. The HR decreases when one of the two variables declines. Despite this, the results show that even with only 30 % of V2G acceptance, 15 million EV providers and 50 % battery dedicated to V2G service, the EVs can provide an energy flow of 34.3 TWh. To be compared to only 0.31 TWh from 5.35 GW of PHS, which currently is present in Germany. Another important benefit of balancing intermittent energy using EVs and V2G is that the service is assumed to be homogeneously spread over the entire country.

While with 50 % V2G acceptance and 50 % EV battery availability, the model shows an HR of 91.9 % with an EV energy flow of 42.6 TWh. These results reflect that V2G can offer significant storage for grid balancing purposes, even without full V2G participation and EV battery availability. Dedicating 50 % of the battery, the users of modern EVs still have at least 150 km of range. In this scenario, EV users must go through the equivalent of 58 full charging cycles annually for the V2G service. If an EV covers 49 km/day at 200 Wh/km, the annual energy consumption is 3577 kWh, which corresponds to 45 charging cycles. With Germany's average vehicle lifespan of 10 years [54], the total charge and discharge cycles combining V2G service and driving would be 1030 cycles. With a total battery life of 3000 charging cycles, 1030 cycles are less than 40 % of its lifetime [55]. As EV batteries have a warranty of 8–10 years and a lifespan of 10 years, the above estimation indicates that V2G will not reduce EV battery life to less than ten years. This implies that an EV user will never have to replace its battery during the entire EV life cycle because of V2G participation. Nevertheless, monetary compensation is likely necessary to achieve high V2G participation from EV users. From the sensitivity analysis, for 100 % SS points, more RES installation puts less stress on EV energy flow and reduces the charging cycles to 29 cycles per year instead of 58 cycles with 99 % SS. This reduces the total charging cycles over the life from 1030 to 740 cycles. While reducing SS shows an inverse effect, where reducing to 95 % and 90 % SS points increases the charging cycles to 64 and 80 cycles.

In section three of the results, we present the impact of EV chargers and the EV adoption level in the fleet on V2G energy flow and grid HR. This study focuses on co-relating charging power with battery life. Considering battery chemical topology, charging batteries at high Crates can reduce their life, due to higher induced battery stress with more chemical and thermal reactions. From the results in Section 4.3, we can observe that fast DC chargers are not required for V2G purposes. The results show that an 11 kW AC charger gives the same HR, energy flow through EV and the number of charging cycles as a 100 kW DC charger. Using an 11-kW charger for a 50-kWh battery pack, the charging occurs at less than 0.5 C. This suggests that an 11-kW charger would be enough to provide the V2G service, without compromising the battery life. Studies show that slow charging with 0.5 C, actually can have a positive influence on battery life [56]. This certainly provides positive confidence for EV users to actively participate in V2G service. For 100 % SS, the EV energy flow decreases compared to other SS points along with the number of charging cycles. This is because at 100 % SS, high RE installation supplies more electricity while the EV contributes a small proportion each hour. Smaller contributions from EV batteries leave space for smaller energy storage during high electricity generation periods, which leads to lower electricity flow between EVs and the grid. With reduced SS EVs provide more electricity via V2G, leaving space for more energy to be stored. This increases the electricity flow and the charging cycles at lower SS points.

While comparison with different EV adoption levels shows that more EVs will not necessarily be an advantage for the V2G purpose and not to the electricity grid. More EVs demand more electric energy, which reduces HR. Even though the system SS goes down, however, having more EVs can reduce the need for fossil-fuel vehicles and thus gasoline use and associated greenhouse gas emissions. Having more EVs instead of gasoline vehicles has environmental and social benefits, which are not considered in this study. Even though this conceptual case study is performed based on German data, the V2G technology could be a good solution for many countries around the world. The required RES installation and thus the potential contribution of V2G to grid balancing changes based on geographical location and RES generation potential. V2G could make a significant contribution in any country with large EV fleets and high RES installed capacity. Nevertheless, the outcome cannot just be extrapolated for certain countries with different geographies and climates. For example, in countries with a cold climate, the EV range and capacity drastically reduce by 25%-40 % during winter times [57]. This will reduce the HR and system SS of the grid. Such operational conditions need to be assessed separately.

#### 5.1. Model limitations

Comparing the model from the study with the real world, our model assumes a centralised V2G system that controls the energy transfers. In the real world, a centralised system has not yet been implemented. Instead, uncontrolled EV charging is still in practice. Uncontrolled energy transfer can result in voltage and frequency fluctuation in the grid which can further lead to breakouts on a large scale [58,59]. Through a centralised control system, it would be possible to assess the SOC of EV batteries and perform energy transfer simultaneously. We can also avoid vehicles with SOC less than minimum SOC during energy extraction and vehicles with SOC higher than maximum SOC during the charging process. Even though at the current stage this can be considered a model limitation, a centralised control system for EVs is essential in the future. Further, the model has a few more simplifying assumptions.

- 1. Choosing V2G as the primary ESS
- 2. Having the same EV moving distribution throughout the year
- 3. Considering the SOC of EVs together as a big battery pack
- 4. Whenever an EV is parked, it is connected to a bidirectional charger
- 5. Considering Germany as an Island grid

In our study, the EVs act as the primary ESS whereas in real life other storage options, e.g., PHES, may be more economical in certain instances. Our model does not take that into account but rather shows the maximum potential of V2G technology. If the model would simulate each individual EV and battery pack, the simulation would become computationally too demanding. To overcome this, future work will address these assumptions with further micro-modelling and clustering which consider vehicles with multiple user profiles and charging patterns. Through clustering, we can consider different vehicle clusters based on SOC and moving patterns, as in the study by Sagaria et al. [43]. Considering the assumption that EVs are always connected to a bidirectional charger, whenever parked, the model assumes a 1:1 charger to EVs ratio. This assumption is very optimistic with current developments. However, sustainable development in the transportation sector shows exponential growth in charging stations and with proper infrastructure planning could pave the way for more efficient and scalable charging solutions. Furthermore, advancements in the field of wireless charging are expected to simplify the challenges associated with connectivity. This study also explores the impact of low charger-to-EV ratios by modelling different vehicle-to-grid (V2G) acceptance rates, which can be interpreted as reflecting varying availability rates of bidirectional chargers in-directly. This analysis helps in understanding the practical implications of V2G systems under different infrastructure scenarios, enhancing the model's relevance and applicability in real-world settings. Finally, the study considers Germany as an Island grid. Being a net positive energy exporter of electricity in 2020 and 2021 [60], the generated excess energy can be used for energy exchange between other countries. The influence of energy import and export also helps to reduce RE installations.

Furthermore, the current model does not consider the degradation of batteries participating in V2G and its economic implications. Participating in V2G increases the number of cycles EV goes through a year by a factor of 2 or 3, which reduces the battery life. A study by Thingvad et al. [61] shows a battery degradation of 10 % and 17.8 % for a period of 2 vears and 5 years with a 23 kWh battery delivering primary frequency regulation for 15 h per day with a daily energy throughput of 50.6 kWh respectively. Analysing different battery charging strategies for EVs by Bui et al. [62], the study reports a battery degradation of 0.0165 % with V2G against 0.0140 % for a week of EV usage. The additional degradation of 0.0025 %/week of capacity fade is experienced by the battery due to V2G technology, which is 17 % higher than normal battery degradation. These study shows that V2G surely increase the battery degradation process, which adversely affects the V2G acceptance rate. However, offering compensation may help to offset this negative effect of V2G service. In future studies, the authors aim to overcome the simplistic assumptions and limitations, which are further discussed in section 5.2.

# 5.2. Future studies

Through future studies, the authors intend to address and discuss the limitations of the current model and improve the model with suitable measures. We aim to refine the results from the model by concentrating on EV clusters with similar usage profiles and charging behaviours through micro-modelling and clustering techniques instead of the whole EV fleet. Specifically, clustering based on average vehicle movement patterns for regular weekdays, weekends, and seasonal variations will enable detailed examining different EV movement distributions that impact the V2G energy flow throughout the year. Future work will also focus to over the assumption of Germany as an isolated grid. Future work will include the possibilities of energy import and export during energy excess and shortage points to limit renewable energy installations. Furthermore, we intend to include flexibilities to energy sources and energy pricing for energy generation options. Considering that all nations pursue cheaper energy, the price of electricity from different sources in the electricity market plays a pivotal role in the acceptance of V2G technology. On top of that, future studies will explore the role of V2G in supporting the grid with lower geographical resolution, representing multiple states and zones in a country. Each node/ state includes energy generation and consumption units and the simulation will show the energy flow between the EV and grid at a microscopic level (within the zones and states) and a macroscopic level (between zones and states). Finally, the authors aim to develop a battery degradation model to examine the degradation in EVs caused by V2G at a national and individual level. The estimation of battery degradation helps to determine the loss of opportunity for EV owners in terms of range which leads to the estimation of V2G compensation cost for EV owners.

## 6. Conclusion

This study examines the social and technical variables that influence the adoption of V2G technology in a RES-supported grid system. The model developed simulates annual energy demand and supply scenarios with the 1-h resolution, considering PV, wind, and hydropower as RES, and EV and PHS as ESS. This study comprehensively examines the influence of V2G acceptance rate, EV EV battery availability, EV charger power and EV volume in a grid supported by V2G technology and RES.

The primary simulations predict that in Germany, by 2030, 71 % of electricity could be generated by PV and wind, with 190 GW of PV and 170 GW of wind turbine installations combined with V2G for storage purposes, providing 99 % self-sufficiency (SS). For a 100 % SS, an additional 90 GW of wind power is required compared to 99 % SS to meet the peak demand points. Simulation results also show that low V2G acceptance and EV battery availability have positive outcomes. With a 30 % V2G acceptance and dedicating half of the EV battery capacity to V2G, the hourly reliability (HR) can reach 86.6 %, significantly higher than the 72 % achieved with zero V2G acceptance and no dedicated EV battery capacity for V2G. Increasing V2G acceptance to 50 % enhances the HR further by 5.3 %–91.9 %. Analysis of the required charger power and the level of EV adoption suggests that an AC charger capacity of 7 kW or 11 kW is sufficient to support V2G services and high-power bidirectionalDC chargers are not essential. The results for 2030 indicate that 15 million EVs could provide substantial energy storage opportunities, facilitating higher penetration of RES without the need for additional ESS.

The results indicate that with 10 million EVs, over 90 % HR is achievable with 50 % V2G acceptance and dedicating 50 % of the EV battery capacity for V2G, using a 7-kW charger. This scenario demonstrates that not all EV owners need to engage in V2G services to significantly balance intermittent energy generation and distribution in the grid. Allocating 50 % of the battery for V2G would also create 250 GWh of energy storage capacity. In conclusion, this paper endorses V2G technology as a viable ESS for future large-scale renewable electricity systems, suggesting that V2G can conserve financial and material resources for other sustainable development initiatives in society.

#### CRediT authorship contribution statement

Shemin Sagaria: Conceptualization, Formal analysis, Investigation, Methodology, Software, Visualization, Writing – original draft. Mart van der Kam: Conceptualization, Methodology, Supervision, Visualization, Writing – review & editing. Tobias Boström: Conceptualization, Methodology, Supervision, Visualization, Writing – review & editing.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Data availability

Data will be made available on request.

## Acknowledgement

The authors would like to acknowledge the Arctic Centre for Sustainable Energy at the UiT - Arctic University of Norway for partial funding of this work.

#### APPENDIX

Appendix 1 (A1): Simulation results from PV analysis.

#### Appendix Table 1

#### Simulation results from PV analysis

Self-su	ifficiency	r: 100 %	Self-su	ifficiency	y: 99 %	Self-su	ifficiency	r <b>: 97</b> %	Self-su	ifficiency	7: 95 %	Self-su	ifficiency	r: 92 %	Self-su	ifficiency	r: 90 %
PV (GW)	Wind (GW)	Excess energy (TWh)															
100	266.3	232.8	100	251.3	204.3	100	229.0	162.8	100	210.5	130.2	100	191.4	99.0	100	179.1	80.1
115	262.0	236.2	115	233.5	181.1	115	211.7	140.8	115	196.7	115.0	115	177.7	84.7	115	166.4	67.5
130	256.5	237.3	130	218.5	163.9	130	195.1	120.6	130	181.9	98.8	130	164.9	72.2	130	153.9	55.3
145	252.1	241.4	145	203.2	145.7	145	180.9	105.7	145	169.0	86.0	145	151.9	59.7	145	142.7	45.9
160	248.2	246.7	160	189.6	132.5	160	169.2	95.6	160	155.5	72.6	160	140.0	49.4	160	132.0	38.1
175	244.7	253.1	175	179.5	125.9	175	158.2	87.0	175	144.5	64.2	175	130.0	42.7	175	123.3	33.6
190	241.7	260.6	190	170.3	121.2	190	149.8	83.9	190	134.9	59.1	190	122.1	39.8	190	114.1	29.1
205	239.0	268.6	205	163.8	121.7	205	143.0	83.8	205	129.4	61.4	205	115.3	40.1	205	108.1	30.7
220	236.6	277.2	220	158.1	123.7	220	137.2	85.9	220	124.4	64.3	220	110.2	43.2	220	103.2	34.5
235	234.1	285.8	235	154.1	129.2	235	133.1	91.4	235	120.3	69.6	235	106.1	48.2	235	99.6	40.4
250	231.6	294.5	250	149.4	133.3	250	129.7	98.3	250	116.6	76.3	250	101.8	53.9	250	95.3	45.7
265	229.6	304.2	265	145.4	139.1	265	127.3	107.0	265	113.7	84.1	265	99.3	62.6	265	92.7	54.1
280	227.8	314.5	280	142.3	146.7	280	124.7	115.4	280	111.3	92.8	280	96.9	71.5	280	90.4	63.1
295	226.1	321.9	295	139.4	154.7	295	122.3	124.4	295	108.7	101.5	295	94.5	80.2	295	87.7	71.9
310	220.3	329.1	310	136.8	163.1	310	120.5	134.4	310	107.0	111.6	310	92.0	89.0	310	85.9	81.6
325	215.6	331.3	325	134.5	172.5	325	118.2	143.5	325	104.9	121.0	325	89.9	98.7	325	83.6	90.8
340	212.5	338.6	340	131.5	180.3	340	116.3	153.4	340	102.8	130.7	340	88.1	108.7	340	81.5	100.3
355	210.4	348.2	355	129.2	189.1	355	113.6	161.9	355	100.8	140.2	355	86.4	119.0	355	79.7	110.4
370	206.5	354.1	370	127.3	199.1	370	111.0	170.5	370	98.4	149.2	370	84.4	128.5	370	77.7	120.1
385	203.6	362.1	385	124.8	208.1	385	109.5	181.0	385	96.3	158.6	385	82.3	138.0	385	75.8	129.9
400	199.6	367.8	400	123.1	218.5	400	107.7	191.0	400	93.9	167.8	400	80.3	147.6	400	74.0	139.9





Appendix Fig. 1. Simulation results of V2G acceptance vs Charger power.

Appendix 3. (A3): Simulation results of V2G acceptance vs. EV penetration Level



Appendix Fig. 2. Simulation results of V2G acceptance vs. EV penetration Level

Appendix 4. (A4): Simulation result of EV battery availability vs EV penetration level



Appendix Fig. 3. Simulation result of EV battery availability vs EV penetration level.





Appendix Fig. 4. Simulation results of EV battery availability vs Charger power.

## References

- European Union, "2050 long-term strategy." https://ec.europa.eu/clima/eu-actio n/climate-strategies-targets/2050-long-term-strategy\_en.
- [2] European Commission, "A European Green Deal." https://ec.europa.eu/info/strate gy/priorities-2019-2024/european-green-deal\_en.
- [3] Child M, Kemfert C, Bogdanov D, Breyer C. Flexible electricity generation, grid exchange and storage for the transition to a 100% renewable energy system in Europe. Renew Energy 2019;139:80–101. https://doi.org/10.1016/j. renene.2019.02.077.
- [4] Nations United. Renewable energy powering a safer future. Climate Action 2021: 1–7 [Online]. Available: https://www.un.org/en/climatechange/raising-ambiti on/renewable-energy.
- [5] Tan KM, Babu TS, Ramachandaramurthy VK, Kasinathan P, Solanki SG, Raveendran SK. Empowering smart grid: a comprehensive review of energy storage technology and application with renewable energy integration. J Energy Storage 2021;39(February):102591. https://doi.org/10.1016/j.est.2021.102591.

- [6] Hrnčić B, Pfeifer A, Jurić F, Duić N, Ivanović V, Vušanović I. Different investment dynamics in energy transition towards a 100% renewable energy system. Energy 2021;237. https://doi.org/10.1016/j.energy.2021.121526.
- [7] Tetteh S, Yazdani MR, Santasalo-Aarnio A. Cost-effective Electro-Thermal Energy Storage to balance small scale renewable energy systems. J Energy Storage 2021;41 (May):102829. https://doi.org/10.1016/j.est.2021.102829.
- [8] Teki VK, Maharan MK, Panigrahi CK. Study on home energy management system with battery storage for peak load shaving. Mater Today Proc 2019;39:1945–9. https://doi.org/10.1016/j.matpr.2020.08.377.
- [9] Al-Ghussain L, Darwish Ahmad A, Abubaker AM, Mohamed MA. An integrated photovoltaic/wind/biomass and hybrid energy storage systems towards 100% renewable energy microgrids in university campuses. Sustain Energy Technol Assessments 2021;46(May):101273. https://doi.org/10.1016/j.seta.2021.101273.
- [10] Giarola S, et al. The role of energy storage in the uptake of renewable energy: a model comparison approach. Energy Pol 2021;151(February):112159. https://doi. org/10.1016/j.enpol.2021.112159.

- [11] Sagaria S, Boström T. Electric vehicle and vehicle to grid technology influence on renewable energy supported grid – a case study on Germany [Online]. Available: https://ieeexplore.ieee.org/abstract/document/10048075; 2022.
- [12] Sagaria S, Boström T, van der Kam M. Conceptualization of a vehicle to grid assisted renewable energy system in Spain. Energy Convers Manag 2024;X:1–34.
- [13] Sree Lakshmi G, Divya G, Sravani G. V2G Transfer of energy to various applications. E3S Web of Conferences 2019;87(201 9):1–6. https://doi.org/ 10.1051/e3sconf/20198701019.
- [14] Mozafar MR, Amini MH, Moradi MH. Innovative appraisement of smart grid operation considering large-scale integration of electric vehicles enabling V2G and G2V systems. Elec Power Syst Res 2018;154:245–56. https://doi.org/10.1016/j. epsr.2017.08.024.
- [15] Wei L, Yi C, Yun J. Energy drive and management of smart grids with high penetration of renewable sources of wind unit and solar panel. Int J Electr Power Energy Syst 2021;129(January):106846. https://doi.org/10.1016/j. iienes.2021.106846.
- [16] Ben Sassi H, Alaoui C, Errahimi F, Es-Sbai N. Vehicle-to-grid technology and its suitability for the Moroccan national grid. J Energy Storage 2021;33(May 2020). https://doi.org/10.1016/j.est.2020.102023.
- [17] Zheng Y, Shao Z, Jian L. The peak load shaving assessment of developing a useroriented vehicle-to-grid scheme with multiple operation modes: the case study of Shenzhen, China. Sustain Cities Soc 2021;67(January):102744. https://doi.org/ 10.1016/j.scs.2021.102744.
- [18] Moura Po, Pires A, Delgado J, de Almeida AT. Grid to vehicle and vehicle to grid systems for large-scale penetration of renewable generation. Eceee Summer Study Proceedings 2019:1025–34. 2019-June.
- [19] Sagaria S, Duarte G, Neves D, Baptista P. Photovoltaic integrated electric vehicles: assessment of synergies between solar energy, vehicle types and usage patterns. J Clean Prod 2022;348(March):131402. https://doi.org/10.1016/j. iclepro.2022.131402.
- [20] Zhao G, Baker J. Effects on environmental impacts of introducing electric vehicle batteries as storage - a case study of the United Kingdom. Energy Strategy Rev Mar. 2022;40:100819. https://doi.org/10.1016/J.ESR.2022.100819.
- [21] Ramaiah Y, Amrutha K, Satwika SV. DC fast charging architecture used in micro grid for vehicle to grid technology. Journal of Electronics and Communication Systems 2023;8(2):19–32. https://doi.org/10.46610/joecs.2023.v08i02.003.
- [22] Li R, Ren H, Wu Q, Li Q, Gao W. Cooperative economic dispatch of EV-HV coupled electric-hydrogen integrated energy system considering V2G response and carbon trading. Renew Energy 2024;227(October 2023):120488. https://doi.org/ 10.1016/j.renene.2024.120488.
- [23] Boström T, Babar B, Hansen JB, Good C. The pure PV-EV energy system a conceptual study of a nationwide energy system based solely on photovoltaics and electric vehicles. Smart Energy 2021;1:100001. https://doi.org/10.1016/j. seev.2021.100001.
- [24] O'Neill D, Yildiz B, Bilbao JI. An assessment of electric vehicles and vehicle to grid operations for residential microgrids. Energy Rep 2022;8:4104–16. https://doi. org/10.1016/j.egvr.2022.02.302.
- [25] Schuller A, Flath CM, Gottwalt S. Quantifying load flexibility of electric vehicles for renewable energy integration. Appl Energy Aug. 2015;151:335–44. https://doi. org/10.1016/J.APENERGY.2015.04.004.
- [26] Nezamoddini N, Wang Y. Risk management and participation planning of electric vehicles in smart grids for demand response. Energy Dec. 2016;116:836–50. https://doi.org/10.1016/J.ENERGY.2016.10.002.
- [27] Elkholy MH, et al. Techno-economic configuration of a hybrid backup system within a microgrid considering vehicle-to-grid technology: a case study of a remote area. Energy Convers Manag Feb. 2024;301:118032. https://doi.org/10.1016/J. ENCONMAN.2023.118032.
- [28] Bibak B, Tekiner-Mogulkoc H. Influences of vehicle to grid (V2G) on power grid: an analysis by considering associated stochastic parameters explicitly. Sustainable Energy, Grids and Networks 2021;26. https://doi.org/10.1016/j. segan.2020.100429.
- [29] Li X, et al. A cost-benefit analysis of V2G electric vehicles supporting peak shaving in Shanghai. Elec Power Syst Res 2020;179(September 2019):106058. https://doi. org/10.1016/j.epsr.2019.106058.
- [30] Drude L, Pereira Junior LC, Rüther R. Photovoltaics (PV) and electric vehicle-togrid (V2G) strategies for peak demand reduction in urban regions in Brazil in a smart grid environment. Renew Energy 2014;68:443–51. https://doi.org/10.1016/ j.renene.2014.01.049.
- [31] Sovacool BK, Noel L, Axsen J, Kempton W. The neglected social dimensions to a vehicle-to-grid (V2G) transition: a critical and systematic review. Environ Res Lett 2018;13(1). https://doi.org/10.1088/1748-9326/aa9c6d.
- [32] Esmaili M, Shafiee H, Aghaei J. Range anxiety of electric vehicles in energy management of microgrids with controllable loads. J Energy Storage 2018;20 (June):57–66. https://doi.org/10.1016/j.est.2018.08.023.
- [33] Geske J, Schumann D. Willing to participate in vehicle-to-grid (V2G)? Why not. Energy Pol 2018;120(March):392–401. https://doi.org/10.1016/j. enpol.2018.05.004.
- [34] Kester J, Noel L, Zarazua de Rubens G, Sovacool BK. Promoting Vehicle to Grid (V2G) in the Nordic region: expert advice on policy mechanisms for accelerated diffusion. Energy Pol 2018;116(March):422–32. https://doi.org/10.1016/j. enpol.2018.02.024.
- [35] Noel L, Zarazua de Rubens G, Kester J, Sovacool BK. Navigating expert skepticism and consumer distrust: rethinking the barriers to vehicle-to-grid (V2G) in the Nordic region. Transport Pol 2019;76(January):67–77. https://doi.org/10.1016/j. tranpol.2019.02.002.

- [36] van der Kam MJ, Meelen AAH, van Sark WGJHM, Alkemade F. Diffusion of solar photovoltaic systems and electric vehicles among Dutch consumers: implications for the energy transition. Energy Res Social Sci 2018;46(July):68–85. https://doi. org/10.1016/j.erss.2018.06.003.
- [37] Meelen T, Doody B, Schwanen T. Vehicle-to-Grid in the UK fleet market: an analysis of upscaling potential in a changing environment. J Clean Prod 2021;290: 125203. https://doi.org/10.1016/j.jclepro.2020.125203.
- [38] Wind energy Europe, "Hourly electricity mix." https://windeurope.org/about-win d/daily-wind/electricity-mix.
- [39] Bundesregierung, "EEG amendment," [Online]. Available: https://www.bundesre gierung.de/breg-de/themen/klimaschutz/amendment-of-the-renewables-act-2060448.
- [40] Reuters. New German government aims for at least 15 million EVs by 2030. Auto. com. 2021. https://auto.economictimes.indiatimes.com/news/industry/new-ge rman-government-aims-for-at-least-15-million-evs-by-2030/87894302.
- [41] Enerdata, "Germany energy information." https://www.enerdata.net/estore/energy-market/germany/.
- [42] Stromnetz Berlin, "Energy usage profile." https://www.stromnetz.berlin/en.
  [43] Sagaria S, Moreira A, Margarido F, Baptista P. From microcars to heavy-duty
- (43) Sagaria S, Moreira A, Margarido F, Baptista P. From interocars to neavy-duty vehicles: vehicle performance comparison of battery and fuel cell electric vehicles. Vehicles 2021;3(4):691–720. https://doi.org/10.3390/vehicles3040041.
- [44] Kostopoulos ED, Spyropoulos GC, Kaldellis JK. Real-world study for the optimal charging of electric vehicles. Energy Rep 2020;6:418–26. https://doi.org/10.1016/ j.egyr.2019.12.008.
- [45] Statista, "Pure pumped storage capacity in Germany from 2008 to 2020." http s://www.statista.com/statistics/867868/pure-pumped-storage-capacity-in-germ any/.
- [46] Bloomberg, "Germany Brings Forward Goal of 100% Renewable Power to 2035." https://www.bloomberg.com/news/articles/2022-02-28/germany-brings-forwar d-goal-of-100-renewable-energy-to-2035%0A.
- [47] Wind energy Europe, "Hourly energy mix." https://windeurope.org/about-wind/ daily-wind/electricity-mix?utf8=&areas=DE&commit=Apply+filters.
- [48] I. International renewable energy agency, "REmap 2030, Renewable Energy Prospects: Germany." [Online]. Available: https://www.irena.org/-/media/Files/ IRENA/Agency/Publication/2015/IRENA\_REmap\_Germany\_summary\_2015\_EN. PDF?la=en&hash=DBEA29F550223310044433EE3060A79F262165D2.
- [49] Wikipedia. Solar power in Germany. Power; 2007. https://en.wikipedia.org/wiki /Solar power in Germany.
- [50] Valentine S. Wind power in Germany. Wind power politics and policy. 2015. htt ps://en.wikipedia.org/wiki/Wind power in Germany.
- [51] Sovacool BK, Kester J, Noel L, Zarazua de Rubens G. Actors, business models, and innovation activity systems for vehicle-to-grid (V2G) technology: a comprehensive review. Renew Sustain Energy Rev 2020;131(April):109963. https://doi.org/ 10.1016/j.rser.2020.109963.
- [52] Kester J, Zarazua de Rubens G, Sovacool BK, Noel L. Public perceptions of electric vehicles and vehicle-to-grid (V2G): insights from a Nordic focus group study. Transport Res Transport Environ 2019;74(August):277–93. https://doi.org/ 10.1016/j.trd.2019.08.006.
- [53] Clean energy wire, "Germany's 2022 renewables and efficiency reforms." https ://www.cleanenergywire.org/factsheets/germanys-2022-renewables-and-energyreforms.
- [54] ACEA. Average age of the EU vehicle fleet, by country. European Automobile Manufacturers' Association (ACEA); 2022. https://www.acea.auto/figure/average -age-of-eu-vehicle-fleet-by-country/?utm\_source=substack&utm\_medium=email.
- [55] ZIGWHEELS, "EV Tech Explained: Battery Cycles In Electric Vehicles." https:// www.zigwheels.com/news-features/ev-guide/ev-tech-explained-what-are-batterycycles/45591/#:~:text=Upcoming Electric Bikes in India&text=For further knowledge%2C a Lithium,any other type of battery.
- [56] Uddin K, Jackson T, Widanage WD, Chouchelamane G, Jennings PA, Marco J. On the possibility of extending the lifetime of lithium-ion batteries through optimal V2G facilitated by an integrated vehicle and smart-grid system. Energy 2017;133: 710–22. https://doi.org/10.1016/j.energy.2017.04.116.
- [57] Sagaria S, Neto RC, Baptista P. Modelling approach for assessing influential factors for EV energy performance. Sustain Energy Technol Assessments Apr. 2021;44: 100984. https://doi.org/10.1016/J.SETA.2020.100984.
- [58] Gamil MM, Senjyu T, Masrur H, Takahashi H, Lotfy ME. Controlled V2Gs and battery integration into residential microgrids: economic and environmental impacts. Energy Convers Manag Feb. 2022;253:115171. https://doi.org/10.1016/ J.ENCONMAN.2021.115171.
- [59] Dias FG, Scoffield D, Mohanpurkar M, Hovsapian R, Medam A. Impact of controlled and uncontrolled charging of electrical vehicles on a residential distribution grid. International conference on probabilistic methods applied to power systems (PMAPS). 2018. https://ieeexplore.ieee.org/stamp/stamp.jsp? arnumber=8440511. [Accessed 3 January 2024].
- [60] Statista, "Import and export volume of electricity in Germany from 1990 to 2021," [Online]. Available: Finally, the study consider Germany as an Island grid. Being a net positive energy exporter in 2020 and 2021.
- [61] Thingvad A, Calearo L, Andersen PB, Marinelli M. Empirical capacity measurements of electric vehicles subject to battery degradation from V2G services. IEEE Trans Veh Technol 2021;70:7547–757.
- [62] Bui TMN, Sheikh M, Dinh TQ, Gupta A, Widanalage DW, Marco J. A study of reduced battery degradation through state-of-charge pre-conditioning for vehicleto-grid operations. IEEE Access 2021;9 [Online]. Available: https://ieeexplore. ieee.org/abstract/document/9617644.